



**Baishideng
Publishing
Group**

7041 Koll Center Parkway, Suite
160, Pleasanton, CA 94566, USA
Telephone: +1-925-223-8242
E-mail: bpgoffice@wjgnet.com
https://www.wjgnet.com

PEER-REVIEW REPORT

Name of journal: World Journal of Clinical Oncology

Manuscript NO: 47545

Title: Ipsilateral breast tumor recurrence in early stage breast cancer patients treated with breast conserving surgery and adjuvant RT: concordance of biomarkers and tumor location from primary tumor to IBTR

Reviewer's code: 03086178

Position: Editorial Board

Academic degree: MD, PhD

Professional title: Doctor, Professor

Reviewer's country: India

Author's country: United States

Reviewer chosen by: Ying Dou

Reviewer accepted review: 2019-07-25 04:16

Reviewer performed review: 2019-07-25 04:58

Review time: 1 Hour

SPECIFIC COMMENTS TO AUTHORS

COMMENTS: The manuscript is written legibly and is within the scope of journal with following features; Originality: Satisfactory Conciseness and precision: Satisfactory Framing and completeness: Good Scientific completeness: There is clear hypothesis, with scientific justification and good for readers. References: As per journal requirements.

Thank you for commenting that the hypothesis is clear with scientific justification. Thank you for addressing that the references are as per journal requirements.



**Baishideng
Publishing
Group**

7041 Koll Center Parkway, Suite
160, Pleasanton, CA 94566, USA
Telephone: +1-925-223-8242
E-mail: bpgoffice@wjgnet.com
https://www.wjgnet.com

PEER-REVIEW REPORT

Name of journal: World Journal of Clinical Oncology

Manuscript NO: 47545

Title: Ipsilateral breast tumor recurrence in early stage breast cancer patients treated with breast conserving surgery and adjuvant RT: concordance of biomarkers and tumor location from primary tumor to IBTR

Reviewer's code: 03769068

Position: Editorial Board

Academic degree: PhD

Professional title: Professor

Reviewer's country: Brazil

Author's country: United States

Reviewer chosen by: Ying Dou

Reviewer accepted review: 2019-07-19 21:11

Reviewer performed review: 2019-07-26 14:37

Review time: 6 Days and 17 Hours

SPECIFIC COMMENTS TO AUTHORS

The manuscript "Ipsilateral breast tumor recurrence in early stage breast cancer patients treated with breast conserving surgery and adjuvant RT: concordance of biomarkers and tumor location from primary tumor to IBTR" submitted by Purswani et al. is a very nice article from a retrospective study that evaluates the primary and recurrent tumor characteristics in patients who experience an in-breast tumor recurrence. This is a good original article and I suggest its publication in World Journal of Clinical Oncology.

Thank you for commenting that the manuscript is a good original article. Thank you for suggesting its publication in World Journal of Clinical Oncology.



**Baishideng
Publishing
Group**

7041 Koll Center Parkway, Suite
160, Pleasanton, CA 94566, USA
Telephone: +1-925-223-8242
E-mail: bpgoffice@wjgnet.com
https://www.wjgnet.com

PEER-REVIEW REPORT

Name of journal: World Journal of Clinical Oncology

Manuscript NO: 47545

Title: Ipsilateral breast tumor recurrence in early stage breast cancer patients treated with breast conserving surgery and adjuvant RT: concordance of biomarkers and tumor location from primary tumor to IBTR

Reviewer's code: 03670885

Position: Editorial Board

Academic degree: PhD

Professional title: Academic Research, Associate Professor, Research Scientist

Reviewer's country: Mexico

Author's country: United States

Reviewer chosen by: Ying Dou

Reviewer accepted review: 2019-07-18 16:10

Reviewer performed review: 2019-08-05 13:54

Review time: 17 Days and 21 Hours

SPECIFIC COMMENTS TO AUTHORS

Juhi M Purswani and co-authors have submitted an interesting case reports about "Ipsilateral breast tumor recurrence in early stage breast cancer patients treated with breast conserving surgery and adjuvant RT: concordance of biomarkers and tumor location from primary tumor to IBTR". In my personal opinion t is a fluently written manuscript, which provides enough information regarding of the role to choose the nest procedure to treat breast cancer patients, to prevent the neoplasia recurrence.

Thank you for commenting that the manuscript is fluently written and provides information to make decisions in treating breast cancer patients and preventing recurrence.



**Baishideng
Publishing
Group**

7041 Koll Center Parkway, Suite
160, Pleasanton, CA 94566, USA
Telephone: +1-925-223-8242
E-mail: bpgoffice@wjgnet.com
https://www.wjgnet.com

Dear World Journal of Clinical Oncology:

Thank you for returning the revised manuscript. Below are the changes made per the editor's suggestions:

1. Data sharing statement revised.
2. Corresponding author's name, title and detailed address provided.
3. Author affiliations were rearranged with department, university, city, postcode, country, etc.
4. Author contributions were revised according to editor's guidelines.
5. Open access statement was included.
6. All subtitles, first and second, were bolded; second subtitles were also italicized
7. Statistical significance was expressed as $p < 0.05$ in the text.
8. Article highlights section was created including, research background, research motivation, research objective, research methods, research results, research conclusions and research perspectives.
9. References were checked and confirmed that there were no repeated references.
10. PubMed citations numbers (PMID) were added for all references. DOI citations were added to references when available on pub med.
11. All authors were listed on all references .
12. Tables were formatted so that all acronyms were detailed under the tables.
13. Audio core tip was recorded and uploaded.

Please let us know if there are any additional revisions necessary. Thank you.



**Baishideng
Publishing
Group**

7041 Koll Center Parkway, Suite
160, Pleasanton, CA 94566, USA

Telephone: +1-925-223-8242

E-mail: bpgoffice@wjgnet.com

https://www.wjgnet.com

Sincerely,

Naamit Kurshan Gerber, MD, Department of Radiation Oncology, New York University School of
Medicine, 160 E. 34th Street, New York, NY 10016, United States. Naamit.Gerber@nyumc.org

Telephone: +1-212-731-6338

Fax: +1- 212-731-5512